Author’s response to reviews

Title: Role of nurses in medication management at the end of life: a qualitative interview study

Authors:

Bregje Huisman (b.huisman@amsterdamumc.nl)
Eric Geijteman (e.geijteman@erasmusmc.nl)
Marianne Dees (Marianne.Dees@radboudumc.nl)
Noralie Schonewille (noralie@schonewille@gmail.com)
Margriet Wieles (consultatie@kuria.nl)
Lia van Zuyl (c.vanzuylen@erasmusmc.nl)
Karolina Szadek (km.szadek@amsterdamumc.nl)
Agnes van der Heide (a.vanderheide@erasmusmc.nl)

Version: 4 Date: 05 May 2020

Author’s response to reviews:

Editor Comments:

1. Thank you for the previously requested revisions. However we note there is still identifying information in table 1 in the 'Nurse characteristics', 'Family physician characteristics' and 'Medical specialist characteristics'.

   It is BioMed Central policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper (http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-11-9).

Currently, the age and gender information in conjunction with other details, may compromise patient anonymity.

Please amend table 1 to address this issue by removing this information or by providing ages as age-ranges, and removing the age (or sex) information.

Reply: we amended table 1.